Author's response to reviews

Title: Total hospital stay for hip fracture: measuring the variations due to pre-fracture residence, rehabilitation, complications and comorbidities.

Authors:

Anthony W Ireland (adje.ireland@bigpond.com)
Patrick J Kelly (p.kelly@sydney.edu.au)
Robert G Cumming (Robert.cumming@sydney.edu.au)

Version: 4 Date: 16 December 2014

Author's response to reviews:

Dr Christopher Morrey,
for Dr Yun-Hee Jeon
Editor
BMC Health Services Research

Dear Dr Morrey

re: MS: 9719504201282841

It is very good news that the above article is now acceptable for publication. I am proud to have this work published in BMC Health Services Research. In response to the most recent editorial requests please find as follows:

(1) Title Page. This includes:
Title, unchanged from the most recent submitted version.
Total hospital stay for hip fracture: measuring the variations due to pre-fracture residence,
rehabilitation, complications and comorbidities.
Full names, institutional addresses and email addresses for the three authors, and identification of the corresponding author

(2) Author contributions and funding sources: the existing text specifies that no funding was received in respect of the work which generated this paper. (no change)

(3) Acknowledgements: A section has been inserted between Author Contributions and the Competing interests statement. Apart from the officers of Department of Veterans’ Affairs, there were no other technical or academic inputs to this paper.
(4) Formatting: A number of formatting adjustments were made in the prior revision, submitted on 26 November 2014. In the absence of any further specific requests, it is therefore assumed that current format complies with the BMC specifications. Should this not be the case, any alterations will be addressed without delay. (no current changes)

(5) It is understood that there is a standard fee to provide on-line and open access publication. We anticipate receiving your invoice for same at the appropriate time.

Any further questions or concerns regarding this paper will receive prompt attention

With best regards,

Anthony W. Ireland
Corresponding author.
16 December 2014

A new Title page has been requested. In the absence of a specific upload site, this is attached below

All Author information submitted on 26.11.2014 remains correct

TITLE PAGE: manuscript id 9719504201282841

TITLE:

Total hospital stay for hip fracture: measuring the variations due to pre-fracture residence, rehabilitation, complications and comorbidities

Anthony W Ireland, Patrick J Kelly and Robert G Cumming

Anthony W Ireland (Corresponding author).
Department of Veterans’ Affairs, 300 Elizabeth St, Sydney 2000, New South Wales, Australia
School of Public Health, Edward Ford Building, University of Sydney 2006, New South Wales, Australia
email: adje.ireland@bigpond.com

Patrick J Kelly School of Public Health, Edward Ford Building, University of Sydney 2006, New South Wales, Australia
email: p.kelly@sydney.edu.au